Author's response to reviews

Title: Traditional Eye Medicine Use in an Ophthalmic Outpatient Population in South-eastern Nigeria: Socio-demographic and Clinical Correlates.

Authors:

Boniface Ikenna Eze (xy3165767@yahoo.com)
Chimdi Memnofu Chuka-Okosa (chimdi_8691@yahoo.com)
Judith Nkechi Uche (amazingnkechi@yahoo.com)

Version: 2 Date: 22 June 2009

Author's response to reviews:

Recommended revisions in the submitted manuscript.
Re: Traditional Eye Medicine Use in an Ophthalmic Outpatient Population in South-eastern Nigeria: Socio-demographic and Clinical Correlates.

I have carefully affected, as recommended, the under listed format changes in the manuscript.

1. Study design: I have clarified the study design, both in the abstract and the manuscript, by deleting the word ‘prospective’; the design was a comparative cross-sectional survey.
2. Key words: I have deleted the key words from page 3 and pasted same on the title page.
3. Introduction: This has been renamed ‘Background’.
4. Materials & Methods: This has been renamed ‘Methods’
5. Ethical approval: I have included the details of ethical approval by The Ethical Committee of the University of Nigeria Teaching Hospital Enugu.
6. Questionnaire: I have elaborated on the development of the questionnaire, including the pre-test. Furthermore, a copy of same has been uploaded as an additional file.
7. Manuscript sections: I have arranged the manuscript sections in the following order:
   Abstract; Background; Methods; Results; Discussion; Conclusions; Abbreviations; Competing interests; Authors contributions; Acknowledgements; References; Tables; Description of Additional files
8. Competing interests: this has been transferred from the ethics section to its appropriate position between ‘Conclusions’ and Authors’ contributions.
9. References: I have change the format of citing references from numbers in superscript to consecutive numbers in brackets within the text.

Thank You
Dr B I Eze
Corresponding author